Author's response to reviews

Title: The Favorable Kinetics and Balance of Nebivolol-Stimulated Nitric Oxide and Peroxynitrite Release in Human Endothelial Cells

Authors:

R. Preston Mason (rpmason@elucidaresearch.com)
Robert F Jacob (rjacob@elucidaresearch.com)
J. Jose Corbalan (corbalan@ohio.edu)
Damian Szczesny (dszczesny@gumed.edu.pl)
Kinga Matysiak (k.matysiak@ior.poznan.pl)
Tadeusz Malinski (malinski@ohio.edu)

Version: 2 Date: 17 September 2013

Author's response to reviews: see over
Dear Editor,

Attached is our corrected manuscript, “The Favorable Kinetics and Balance of Nebivolol-Stimulated Nitric Oxide and Peroxynitrite Release in Human Endothelial Cells”. We would like to thank you and the reviewers for the very helpful comments and suggestions. All of the suggestions were implemented into the manuscript.

Specifically:
Editor and Reviewer 1 – We included a paragraph indicating study limitations as well as an ‘Acknowledgements’ section.

Reviewer 2 – n-values were added to the figure legends and the statistical significance was included on Figure 4A and 5C.

Thank you for your prompt handling of this manuscript.

With Best Regards,

Tadeusz Malinski, PhD, Dr. h.c.m.
Marvin & Ann Dilley White Chair and Distinguished Professor